\xname{H2-Air10}
\W\subsubsection{Critical energy vs equivalence ratio; H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Air10.ps,width=6in}}
{\htmlimage{\plotdir/H2-Air10.gif}}
\T\caption{Critical energy vs equivalence ratio; H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Air10.ps}

\T at24 - Table~\ref{at24}~\cite[Guirao (1982)]{c_guir82}
\W\xlink{at24.txt}{\rawdatadir/at24.txt}
\W Guirao (1982) \cite{c_guir82}
T=293 K, P=101.3 kPa

\T mk30 - Table~\ref{mk30}~\cite[Atkinson (1980)]{r_atki80}
\W\xlink{mk30.txt}{\rawdatadir/mk30.txt}
\W Atkinson (1980) \cite{r_atki80}
T=293 K, P=101.3 kPa

\T at34b - Table~\ref{at34b}~\cite[Benedick (1986)]{w_bene86}
\W\xlink{at34b.txt}{\rawdatadir/at34b.txt}
\W Benedick (1986) \cite{w_bene86}
T=293 K, P=83.99 kPa

\T at53 - Table~\ref{at53}~\cite[Makeev (1983)]{v_make83}
\W\xlink{at53.txt}{\rawdatadir/at53.txt}
\W Makeev (1983) \cite{v_make83}
T=293 K, P=101.3 kPa

\T at34c - Table~\ref{at34c}~\cite[Benedick (1986)]{w_bene86}
\W\xlink{at34c.txt}{\rawdatadir/at34c.txt}
\W Benedick (1986) \cite{w_bene86}
T=293 K, P=101.3 kPa

\T\clearpage

\xname{H2-Ox17}
\W\subsubsection{Critical energy vs equivalence ratio; H2-O2, H2-Cl2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox17.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox17.gif}}
\T\caption{Critical energy vs equivalence ratio; H2-O2, H2-Cl2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox17.ps}

\T at67a - Table~\ref{at67a}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at67a.txt}{\rawdatadir/at67a.txt}
\W Matsui (1979) \cite{h_mats79}
T=293 K, P=101.3 kPa, Subcategory=spherical, Fuel=H2, Oxidizer=O2

\T at55a - Table~\ref{at55a}~\cite[Litchfield (1962)]{e_litc62}
\W\xlink{at55a.txt}{\rawdatadir/at55a.txt}
\W Litchfield (1962) \cite{e_litc62}
T=293 K, P=101.3 kPa, Subcategory=spherical, spark, Fuel=H2, Oxidizer=O2

\T at55b - Table~\ref{at55b}~\cite[Litchfield (1962)]{e_litc62}
\W\xlink{at55b.txt}{\rawdatadir/at55b.txt}
\W Litchfield (1962) \cite{e_litc62}
T=293 K, P=101.3 kPa, Subcategory=spherical, exploding wire, Fuel=H2, Oxidizer=O2

\T at181a - Table~\ref{at181a}~\cite[Knystautas (1988)]{r_knys88}
\W\xlink{at181a.txt}{\rawdatadir/at181a.txt}
\W Knystautas (1988) \cite{r_knys88}
T=293 K, P=8 kPa, Subcategory=spherical, spark, Fuel=H2, Oxidizer=Cl2

\T at181b - Table~\ref{at181b}~\cite[Knystautas (1988)]{r_knys88}
\W\xlink{at181b.txt}{\rawdatadir/at181b.txt}
\W Knystautas (1988) \cite{r_knys88}
T=293 K, P=16 kPa, Subcategory=spherical, spark, Fuel=H2, Oxidizer=Cl2

\T\clearpage

\xname{H2-Ox18}
\W\subsubsection{Critical energy vs initial pressure - part 1; H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox18.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox18.gif}}
\T\caption{Critical energy vs initial pressure - part 1; H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox18.ps}

\T mk82a - Table~\ref{mk82a}~\cite[Lee (1977)]{j_lee77}
\W\xlink{mk82a.txt}{\rawdatadir/mk82a.txt}
\W Lee (1977) \cite{j_lee77}
T=293 K, ER=1

\T\clearpage

\xname{H2-Ox19}
\W\subsubsection{Critical energy vs initial pressure - part 2; H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox19.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox19.gif}}
\T\caption{Critical energy vs initial pressure - part 2; H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox19.ps}

\T at49a - Table~\ref{at49a}~\cite[Zitoun (1995)]{r_zito95}
\W\xlink{at49a.txt}{\rawdatadir/at49a.txt}
\W Zitoun (1995) \cite{r_zito95}
T=293 K, ER=1

\T at49b - Table~\ref{at49b}~\cite[Zitoun (1995)]{r_zito95}
\W\xlink{at49b.txt}{\rawdatadir/at49b.txt}
\W Zitoun (1995) \cite{r_zito95}
T=123 K, ER=1

\T\clearpage

\xname{H2-Ox20}
\W\subsubsection{Critical energy vs percent additive - part 1; H2-O2-Additive}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox20.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox20.gif}}
\T\caption{Critical energy vs percent additive - part 1; H2-O2-Additive}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox20.ps}

\T at123a - Table~\ref{at123a}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at123a.txt}{\rawdatadir/at123a.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, 0-5\% CH4

\T at123b - Table~\ref{at123b}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at123b.txt}{\rawdatadir/at123b.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, .7-5\% CH3Cl

\T at123d - Table~\ref{at123d}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at123d.txt}{\rawdatadir/at123d.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, 1-5\% CCl4

\T at123e - Table~\ref{at123e}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at123e.txt}{\rawdatadir/at123e.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, 1-5\% CHCl3

\T at123f - Table~\ref{at123f}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at123f.txt}{\rawdatadir/at123f.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, 5\% Cl2

\T\clearpage

\xname{H2-Ox21}
\W\subsubsection{Critical energy vs percent additive - part 2; H2-O2-Additive}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox21.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox21.gif}}
\T\caption{Critical energy vs percent additive - part 2; H2-O2-Additive}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox21.ps}

\T at124b - Table~\ref{at124b}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at124b.txt}{\rawdatadir/at124b.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, 1-5\% C2H6

\T at124c - Table~\ref{at124c}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at124c.txt}{\rawdatadir/at124c.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, 2\% i-C4H10

\T at124d - Table~\ref{at124d}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at124d.txt}{\rawdatadir/at124d.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, 1-3\% n-C4H10

\T at124e - Table~\ref{at124e}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at124e.txt}{\rawdatadir/at124e.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, 2-4\% C3H8

\T at126a - Table~\ref{at126a}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at126a.txt}{\rawdatadir/at126a.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, .5-1.5\% Isobutene

\T\clearpage

\xname{H2-Ox22}
\W\subsubsection{Critical energy vs percent additive - part 3; H2-O2-Additive}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/H2-Ox22.ps,width=6in}}
{\htmlimage{\plotdir/H2-Ox22.gif}}
\T\caption{Critical energy vs percent additive - part 3; H2-O2-Additive}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/H2-Ox22.ps}

\T at126b - Table~\ref{at126b}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at126b.txt}{\rawdatadir/at126b.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, .5-2\% Trans-butene-2

\T at126c - Table~\ref{at126c}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at126c.txt}{\rawdatadir/at126c.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, 1-2\% Propylene

\T at126d - Table~\ref{at126d}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at126d.txt}{\rawdatadir/at126d.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, 2-3\% Butene-1

\T at126e - Table~\ref{at126e}~\cite[Macek (1963)]{a_mace63}
\W\xlink{at126e.txt}{\rawdatadir/at126e.txt}
\W Macek (1963) \cite{a_mace63}
T=297 K, P=101.3 kPa, ER=0.82, 1-4\% Ethylene

\T\clearpage

\xname{CH4-Air2}
\W\subsubsection{Critical energy vs equivalence ratio; CH4-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Air2.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Air2.gif}}
\T\caption{Critical energy vs equivalence ratio; CH4-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Air2.ps}

\T at177a - Table~\ref{at177a}~\cite[Beeson (1991)]{h_bees91}
\W\xlink{at177a.txt}{\rawdatadir/at177a.txt}
\W Beeson (1991) \cite{h_bees91}
T=293 K, P=101.3 kPa

\T\clearpage

\xname{CH4-Air3}
\W\subsubsection{Critical energy vs equivalence ratio; CH4-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Air3.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Air3.gif}}
\T\caption{Critical energy vs equivalence ratio; CH4-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Air3.ps}

\T at115 - Table~\ref{at115}~\cite[Wolanski (1981)]{p_wola81}
\W\xlink{at115.txt}{\rawdatadir/at115.txt}
\W Wolanski (1981) \cite{p_wola81}
T=293 K, P=101.3 kPa

\T\clearpage

\xname{CH4-Ox8}
\W\subsubsection{Critical energy vs equivalence ratio; CH4-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Ox8.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Ox8.gif}}
\T\caption{Critical energy vs equivalence ratio; CH4-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Ox8.ps}

\T at79a - Table~\ref{at79a}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at79a.txt}{\rawdatadir/at79a.txt}
\W Matsui (1979) \cite{h_mats79}
P=101.3 kPa

\T\clearpage

\xname{CH4-Ox9}
\W\subsubsection{Critical energy vs equivalence ratio; CH4-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Ox9.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Ox9.gif}}
\T\caption{Critical energy vs equivalence ratio; CH4-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Ox9.ps}

\T at195c - Table~\ref{at195c}~\cite[Nicholls (1979)]{j_nich79}
\W\xlink{at195c.txt}{\rawdatadir/at195c.txt}
\W Nicholls (1979) \cite{j_nich79}
P=101.3 kPa

\T at195d - Table~\ref{at195d}~\cite[Aminallah (1993)]{m_amin93}
\W\xlink{at195d.txt}{\rawdatadir/at195d.txt}
\W Aminallah (1993) \cite{m_amin93}
P=100 kPa

\T at195e - Table~\ref{at195e}~\cite[Aminallah (1993)]{m_amin93}
\W\xlink{at195e.txt}{\rawdatadir/at195e.txt}
\W Aminallah (1993) \cite{m_amin93}
P=120 kPa

\T at209a - Table~\ref{at209a}~\cite[Desbordes (1973)]{d_desb73}
\W\xlink{at209a.txt}{\rawdatadir/at209a.txt}
\W Desbordes (1973) \cite{d_desb73}
P=100 kPa

\T\clearpage

\xname{CH4-Ox10}
\W\subsubsection{Critical energy vs percent diluent; CH4-O2-N2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/CH4-Ox10.ps,width=6in}}
{\htmlimage{\plotdir/CH4-Ox10.gif}}
\T\caption{Critical energy vs percent diluent; CH4-O2-N2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/CH4-Ox10.ps}

\T at210a - Table~\ref{at210a}~\cite[Bull (1976)]{d_bull76}
\W\xlink{at210a.txt}{\rawdatadir/at210a.txt}
\W Bull (1976) \cite{d_bull76}


\T at210b - Table~\ref{at210b}~\cite[Kogarko (1965)]{s_koga65}
\W\xlink{at210b.txt}{\rawdatadir/at210b.txt}
\W Kogarko (1965) \cite{s_koga65}


\T\clearpage

\xname{C2H2-Ox8}
\W\subsubsection{Critical energy vs initial pressure; C2H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Ox8.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Ox8.gif}}
\T\caption{Critical energy vs initial pressure; C2H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Ox8.ps}

\T at82b - Table~\ref{at82b}~\cite[Lee (1977)]{j_lee77}
\W\xlink{at82b.txt}{\rawdatadir/at82b.txt}
\W Lee (1977) \cite{j_lee77}


\T at54a - Table~\ref{at54a}~\cite[Vasil'ev (1982)]{a_vasi82}
\W\xlink{at54a.txt}{\rawdatadir/at54a.txt}
\W Vasil'ev (1982) \cite{a_vasi82}


\T\clearpage

\xname{C2H2-Ox9}
\W\subsubsection{Critical energy vs initial pressure; C2H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Ox9.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Ox9.gif}}
\T\caption{Critical energy vs initial pressure; C2H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Ox9.ps}

\T at82c - Table~\ref{at82c}~\cite[Lee (1977)]{j_lee77}
\W\xlink{at82c.txt}{\rawdatadir/at82c.txt}
\W Lee (1977) \cite{j_lee77}


\T\clearpage

\xname{C2H2-Air5}
\W\subsubsection{Critical energy vs initial pressure; C2H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Air5.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Air5.gif}}
\T\caption{Critical energy vs initial pressure; C2H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Air5.ps}

\T at105a - Table~\ref{at105a}~\cite[Vasil'ev (1980)]{a_vasi80}
\W\xlink{at105a.txt}{\rawdatadir/at105a.txt}
\W Vasil'ev (1980) \cite{a_vasi80}


\T\clearpage

\xname{C2H2-Air6}
\W\subsubsection{Critical energy vs equivalence ratio; C2H2-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H2-Air6.ps,width=6in}}
{\htmlimage{\plotdir/C2H2-Air6.gif}}
\T\caption{Critical energy vs equivalence ratio; C2H2-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H2-Air6.ps}

\T at108a - Table~\ref{at108a}~\cite[Freiwald (1962)]{h_frei62}
\W\xlink{at108a.txt}{\rawdatadir/at108a.txt}
\W Freiwald (1962) \cite{h_frei62}


\T at108b - Table~\ref{at108b}~\cite[Kogarko (1965)]{s_koga65}
\W\xlink{at108b.txt}{\rawdatadir/at108b.txt}
\W Kogarko (1965) \cite{s_koga65}


\T\clearpage

\xname{C2H4-Air5}
\W\subsubsection{Critical energy vs equivalence ratio; C2H4-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/C2H4-Air5.ps,width=6in}}
{\htmlimage{\plotdir/C2H4-Air5.gif}}
\T\caption{Critical energy vs equivalence ratio; C2H4-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/C2H4-Air5.ps}

\T at109a - Table~\ref{at109a}~\cite[Bull (1978)]{d_bull78}
\W\xlink{at109a.txt}{\rawdatadir/at109a.txt}
\W Bull (1978) \cite{d_bull78}


\T at109b - Table~\ref{at109b}~\cite[Bull (1979)]{d_bull79a}
\W\xlink{at109b.txt}{\rawdatadir/at109b.txt}
\W Bull (1979) \cite{d_bull79a}


\T at109c - Table~\ref{at109c}~\cite[Hikita (1975)]{t_hiki75}
\W\xlink{at109c.txt}{\rawdatadir/at109c.txt}
\W Hikita (1975) \cite{t_hiki75}


\T at32b - Table~\ref{at32b}~\cite[Murray (1981)]{s_murr81}
\W\xlink{at32b.txt}{\rawdatadir/at32b.txt}
\W Murray (1981) \cite{s_murr81}


\T at84a - Table~\ref{at84a}~\cite[Benedick (1986)]{w_bene86}
\W\xlink{at84a.txt}{\rawdatadir/at84a.txt}
\W Benedick (1986) \cite{w_bene86}


\T\clearpage

\xname{HC-Ox3}
\W\subsubsection{Critical energy vs equivalence ratio; C2H6-O2, C2H4-O2, C2H2-O2}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Ox3.ps,width=6in}}
{\htmlimage{\plotdir/HC-Ox3.gif}}
\T\caption{Critical energy vs equivalence ratio; C2H6-O2, C2H4-O2, C2H2-O2}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Ox3.ps}

\T at79b - Table~\ref{at79b}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at79b.txt}{\rawdatadir/at79b.txt}
\W Matsui (1979) \cite{h_mats79}
Fuel=C2H4

\T at44c - Table~\ref{at44c}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at44c.txt}{\rawdatadir/at44c.txt}
\W Matsui (1979) \cite{h_mats79}
Fuel=C2H6

\T at44a - Table~\ref{at44a}~\cite[Matsui (1979)]{h_mats79}
\W\xlink{at44a.txt}{\rawdatadir/at44a.txt}
\W Matsui (1979) \cite{h_mats79}
Fuel=C2H2

\T\clearpage

\xname{HC-Air8}
\W\subsubsection{Critical energy vs equivalence ratio; C2H6-Air, C3H8-Air, C4H10-Air}
\begin{figure}[!ht]
\texorhtml{\epsfig{file=\plotdir/HC-Air8.ps,width=6in}}
{\htmlimage{\plotdir/HC-Air8.gif}}
\T\caption{Critical energy vs equivalence ratio; C2H6-Air, C3H8-Air, C4H10-Air}
\end{figure}

\W\xlink{Postscript Version}{\plotdir/HC-Air8.ps}

\T at32e - Table~\ref{at32e}~\cite[Knystautas (1984)]{r_knys84}
\W\xlink{at32e.txt}{\rawdatadir/at32e.txt}
\W Knystautas (1984) \cite{r_knys84}
Fuel=C2H6

\T at32f - Table~\ref{at32f}~\cite[Elsworth (1984)]{j_elsw84}
\W\xlink{at32f.txt}{\rawdatadir/at32f.txt}
\W Elsworth (1984) \cite{j_elsw84}
Fuel=C3H8

\T at32g - Table~\ref{at32g}~\cite[Elsworth (1984)]{j_elsw84}
\W\xlink{at32g.txt}{\rawdatadir/at32g.txt}
\W Elsworth (1984) \cite{j_elsw84}
Fuel=C4H10

\T at92 - Table~\ref{at92}~\cite[Bull (1982)]{d_bull82a}
\W\xlink{at92.txt}{\rawdatadir/at92.txt}
\W Bull (1982) \cite{d_bull82a}
Fuel=C2H6

\T\clearpage

